Author's response to reviews

Title: An adult patient presenting with appendix perforation in appendix duplication: a case report

Authors:

Emel Canbay (dremelcanbay@gmail.com)
Emel Akman (emelakman1@hotmail.com)

Version: 11 Date: 9 December 2010

Author's response to reviews: see over
Dear Mr Ed Bautista,

Thank you very much for your letter. We have completed last checking before publication. We have changed desired revisions listed below. I confirm that this the last version before going to publish. Thank you for your efforts and helps. We are looking forward to hearing from you soon.

Sincerely yours

Emel Canbay MD, PhD

Corresponding author
Major Changes were made in the MS

1. Consent section and required sentences were added.
2. Competing interests and explanations were added.
3. Authors’ contributions were added.

Minor Changes were made in the MS

1. Visible vertical lines were removed from the table and legend was placed above the table
2. Figure was cut as possible as minimalized.
3. Typography was checked lastly.